Author’s response to reviews

Title: What is the Influence of Tonsillectomy on the Level of Periodontal Pathogens on the Tongue Dorsum and in Periodontal Pockets

Authors:

Valeria Diener (valeria.diener@zzm.uzh.ch)
Annietta Gay (annietta.gay@gmail.com)
Michael Soyka (michael.soyka@usz.ch)
Thomas Attin (thomas.attin@zzm.uzh.ch)
Patrick Schmidlin (patrick.schmidlin@zzm.uzh.ch)
Philipp Sahrmann (philipp.sahrmann@zzm.uzh.ch)

Version: 1 Date: 27 Feb 2018

Author’s response to reviews:

Dear Sir or Madam,

we have registered our study retrospectively in the German Clinical Trials Register (DRK000140778), removed the figures in the text document and added the regarding information for the "declaration section", before we finally removed the tracked changes.

I hope, this was done to your full satisfaction.

Best greetings from Zurich,

Philipp Sahrmann